Author's response to reviews

Title: Genetic mutations in acute pancreatitis

Authors:

Dorota Koziel (dorota.koziel@wp.pl)
Stanislaw Gluszek (sgluszek@wp.pl)
Artur Kowalik (artur_kowalik@yahoo.com)
Małgorzata Chlopek (malgo.chlopek@gmail.com)
Liliana Pieciak (lilianamazurek@o2.pl)

Version: 2

Date: 10 September 2014

Author's response to reviews:

Kielce, 10th September 2014

Dear Editor,

We are pleased send to You our manuscript titled “Genetic mutations in acute pancreatitis” by Koziel D et al. Hereby we submit enclosed paper for consideration for publication in BMC cancer.

According to our knowledge this is the first report showing the connection of mutation SPINK1, CFTR, CTR with the clinic course of pancreatitis acute, etiology of disease, recurrences of illness and other environmental factors known as determinants of acute pancreatitis. The authors of manuscripts published the results of studies of mutations in European Pancreatic Club in 2013 and 2014. Moreover the abbreviation of studies were presented in Pancreatology.

We believe our findings would appeal to the readership of BMC cancer.

Please address all correspondence to:

Dorota Kozie#
Al. IX Wieków Kielc 19
25-317 Kielce
POLAND
Email: dorota.koziel@wp.pl

We look forward to hearing from you at your earliest convenience.

We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal.

All authors have approved the manuscript and agree with its submission to BMC cancer.

Yours Faith fully,

Dorota Kozie#,
Stanisław Guszek,
Artur Kowalik,
Małgorzata Chopek,
Liliana Pięciak